University of Toronto at Scarborough CSC B63H Computer Science Summer 2008

Assignment #

Cover Page

Please fill out this page and attach it (with a staple) to the front of your assignment.

Student Number: E-mail:

Last (Family) Name(s):

First (Given) Name(s):

By virtue of submitting this homework, I acknowledge that I am aware of the policy on homework collaboration for this course. I certify that this homework was completed in accordance with this policy and the University policy on plagiarism.